SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

(MARK ONE)

(X) QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d)

OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 1997

0R

(_) TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d)
OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period fromto...... Commission file number 0-10454

UNIVERSAL HEALTH SERVICES, INC.

(Exact name of registrant as specified in its charter)

DELAWARE 23-2077891

(State or other jurisdiction of Incorporation or Organization)

(I.R.S. Employer Identification No.)

UNIVERSAL CORPORATE CENTER 367 SOUTH GULPH ROAD KING OF PRUSSIA, PENNSYLVANIA 19406

(Address of principal executive office) (Zip Code)

Registrant's telephone number, including area code (610) 768-3300

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes X No _____

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date. Common shares outstanding, as of April 30, 1997:

Class A 2,060,929 Class B 30,022,477 Class C 207,230 Class D 33,954

Page One of Twelve Pages

UNIVERSAL HEALTH SERVICES, INC.

$I\ N\ D\ E\ X$

PART I. FINANCIAL INFORMATION	PAGE NO.
Item 1. Financial Statements	
Consolidated Statements of Income - Three Months Ended March 31, 1997 and 1996	Three
Condensed Consolidated Balance Sheets - March 31, 1997 and December 31, 1996	Four
Condensed Consolidated Statements of Cash Flows Three Months Ended March 31, 1997 and 1996	Five
Notes to Condensed Consolidated Financial Statements	Six & Seven
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	Eight, Nine & Ten
PART II. OTHER INFORMATION	Eleven
SIGNATURE	Twelve

Page Two of Twelve Pages

PART I. FINANCIAL INFORMATION

UNIVERSAL HEALTH SERVICES, INC. AND SUBSIDIARIES
CONSOLIDATED STATEMENTS OF INCOME
(000s omitted except per share amounts)
(unaudited)

	THREE MONTHS ENDED MARCH 31,	
	1997	1996
Net revenues	\$340,170	\$266,523
Operating charges: Operating expenses Salaries and wages Provision for doubtful accounts Depreciation and amortization Lease and rental expense Interest expense, net	129,674 119,747 23,663 19,028 9,121 4,956	102,335 94,500 16,674 14,783 9,405 4,648
	306,189 	242,345
Income before income taxes Provision for income taxes	,	24,178 8,677
NET INCOME	\$ 21,530 ======	\$ 15,501 ======
Earnings per common and common share equivalents:	\$ 0.65 =====	\$ 0.54 ======
Weighted average number of common shares and equivalents:	32,986 =====	28,712 ======

See accompanying notes to these condensed consolidated financial statements.

Page Three of Twelve Pages

UNIVERSAL HEALTH SERVICES, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (000s omitted)

	MARCH 31,	DECEMBER 31,
	1997	1996
	(UNAUDITED)	
ASSETS		
CURRENT ASSETS: Cash and cash equivalents Accounts receivable, net Supplies Deferred income taxes	\$ 2,281 154,952 21,915 9,809	\$ 288 145,364 22,019 12,313
Other current assets	14,414	13,969
Total current assets	203,371	193,953
Property and equipment Less: accumulated depreciation	870,620 (286,761) 583,859	839,564 (271,936) 567,628
OTHER ASSETS:		
Excess of cost over fair value of net assets acquired Deferred income taxes Deferred charges Other	146,491 11,284 10,804 34,154 202,733 \$ 989,963	150,336 9,993 11,237 32,648 204,214 \$ 965,795 =======
LIABILITIES AND STOCKHOLDERS' EQUITY		
CURRENT LIABILITIES: Current maturities of long-term debt Accounts payable and accrued liabilities Federal and state taxes	\$ 6,319 144,752 10,113	\$ 6,866 132,441 772
Total current liabilities	161,184	140,079
Other noncurrent liabilities	91,967	97,102
Long-term debt, net of current maturities	258,693	275,634
COMMON STOCKHOLDERS' EQUITY: Class A Common Stock, 2,060,929 shares outstanding in 1997, 2,060,929 in 1996	21	21
Class B Common Stock, 29,986,504 shares outstanding in 1997, 29,816,153 in 1996	300	298
Class C Common Stock, 207,230 shares outstanding in 1997, 207,230 in 1996	2	2
Class D Common Stock, 34,216 shares outstanding in 1997, 36,805 in 1996 Capital in excess of par, net of deferred compensation of \$460,000 in 1997		
and \$377,000 in 1996 Retained earnings	197,915 279,881	194,308 258,351
	478,119	452,980
	\$ 989,963 =======	\$ 965,795 =======

See accompanying notes to these condensed consolidated financial statements.

UNIVERSAL HEALTH SERVICES, INC. AND SUBSIDIARIES

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (000s omitted)

THREE MONTHS ENDED

	MARCH 31,	
		1996
CASH FLOWS FROM OPERATING ACTIVITIES:	ቀ 31 E30	# 15 501
Net income Adjustments to reconcile net income to net cash provided by operating activities:	\$ 21,530	\$ 15,501
Depreciation & amortization	19,028	14,783
Provision for self-insurance reserves Changes in assets & liabilities, net of effects from acquisitions and dispositions:	4,516	3,044
Accounts receivable	(9.588)	(2.237)
Accrued interest	(3,159)	(3,447)
Accrued and deferred income taxes	12.070	8,117
Other working capital accounts	7,916	(2,237) (3,447) 8,117 3,810
Other assets and deferred charges Other	983	(3,377)
Payments made in settlement of self-insurance claims	(2,354)	(4,314)
NET CASH PROVIDED BY OPERATING ACTIVITIES	49,013	32,681
24011 51 0110 50011 5111/5055110 40551/55550		
CASH FLOWS FROM INVESTING ACTIVITIES:	(00 550)	(05.700)
Property and equipment additions, net	(30,558)	(25,729)
NET CASH USED IN INVESTING ACTIVITIES		
NET CASH USED IN INVESTING ACTIVITIES	(30,336)	(25,729)
CASH FLOWS FROM FINANCING ACTIVITIES:		
Reduction of long-term debt	(17,488)	(6,830)
Issuance of common stock	1 10/1	hiin
NET CASH USED IN FINANCING ACTIVITIES	(16,462)	(6,224)
INCREASE IN CASH AND CASH EQUIVALENTS	1,993	728
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	288	728 34 ======
CACH AND CACH FOUTVALENTS. FND OF BEDTOR	====== \$ 2,281	=======
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 2,281 ======	Φ 10Z
SUPPLEMENTAL DISCLOSURES OF CASH FLOW INFORMATION:		
Interest paid	\$ 8,115	\$ 8,095
	=======	=======
Income toyon poid not of refunda	Ф 004	ф 700
Income taxes paid, net of refunds	\$ 381 ======	\$ 782
	=======	=======

See accompanying notes to these condensed consolidated financial statements.

Page Five of Twelve Pages

UNIVERSAL HEALTH SERVICES, INC. AND SUBSIDIARIES

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(1) GENERAL

The consolidated financial statements included herein have been prepared by the Company, without audit, pursuant to the rules and regulations of the Securities and Exchange Commission and reflect all adjustments which, in the opinion of the Company, are necessary to fairly present results for the interim periods. Certain information and footnote disclosures normally included in financial statements prepared in accordance with generally accepted accounting principles have been condensed or omitted pursuant to such rules and regulations, although the Company believes that the accompanying disclosures are adequate to make the information presented not misleading. It is suggested that these financial statements be read in conjunction with the financial statements, accounting policies and the notes thereto included in the Company's Annual Report on Form 10-K for the year ended December 31, 1996.

Prior to 1997, the Company included charity care services as a component of its provision for doubtful accounts. Effective January 1, 1997, in accordance with health care industry practice, the Company began excluding charity care from net revenues, and has reclassified the 1996 amounts to conform with this presentation. The change in presentation has no effect on reported net income.

(2) EARNINGS PER SHARE

Earnings per share are based on the weighted average number of common shares outstanding during the year adjusted to give effect to common stock equivalents. In April 1996, the Company declared a two-for-one stock split in the form of a 100% stock dividend which was paid in May, 1996. All classes of common stock participated on a pro rata basis. The weighted average number of common shares and equivalents and earnings per common and common equivalent share for the three months ended March 31, 1996 have been adjusted to reflect the two-for-one stock split.

The Financial Accounting Standards Board recently issued Statement 128, Earnings per Share, which is effective for financial statements for periods ending after December 15, 1997. Pursuant to the provisions of Statement 128, the Company's basic earnings per share would have been \$.67 and \$.56 for the three month periods ended March 31, 1997 and 1996, respectively, and the diluted earnings per share would have been \$.65 and \$.54 for the three month periods ended March 31, 1997 and 1996, respectively.

(3) OTHER LIABILITIES

Other noncurrent liabilities include the long-term portion of the Company's professional and general liability and workers' compensation reserves.

(4) COMMITMENT AND CONTINGENCIES

Under certain agreements, the Company has committed or guaranteed an aggregate of \$13 million related principally to the Company's self-insurance programs and as support for various debt instruments and loan guarantees.

Page Six of Twelve Pages

(5) SUBSEQUENT EVENTS

Subsequent to the end of the 1997 first quarter, the Company executed a joint venture partnership agreement, subject to regulatory approval, for the ownership and operation of The George Washington University Hospital, a 501-bed acute care facility located in Washington, D.C. The partnership has recently filed for certificate of need approval. The Company also entered into a management agreement, which commenced in April 1997, to manage the operations of the hospital. Pursuant to the terms of the partnership agreement, the Company will provide an immediate commitment of \$80 million (\$40 million in cash and a \$40 million letter of credit) as part of a total intended investment by the partnership of \$125 million over the next ten years for enhancement of the hospital's operations. The Company will hold an 80% interest in the partnership and The George Washington University will hold a 20% interest.

Page Seven of Twelve Pages

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

GENERAL

The matters discussed in this report as well as the news releases issued from time to time by the Company contain certain forward-looking statements that involve risks and uncertainties, including, among other things, that the majority of the Company's revenues are produced by a small number of its total facilities, possible changes in levels and terms of reimbursement for the Company's charges by government programs or other third party payors, the ability of the Company to successfully integrate its recent and proposed acquisitions and the ability to continue to finance growth on favorable terms.

RESULTS OF OPERATIONS

Net revenues increased 28% or \$74 million for the three months ended March 31, 1997 over the comparable prior year period due primarily to the acquisitions of a 357-bed medical complex located in Amarillo, Texas and four behavioral health centers located in Pennsylvania, all of which were acquired during the second quarter of 1996. Also contributing to the increase in net revenues for the three months ended March 31, 1997 as compared to the comparable prior year quarter was a 6% or \$14 million increase in net revenues at hospital facilities owned during both periods.

Earnings before interest, income taxes, depreciation, amortization and lease and rental expense (EBITDAR) increased 27% or \$14 million to \$67 million for the three months ended March 31, 1997 as compared to \$53 million in the comparable prior year period. Overall operating margins were 19.7% and 19.9% for the three months ended March 31, 1997 and 1996, respectively.

ACUTE CARE SERVICES

Net revenues from the Company's acute care hospitals, ambulatory treatment centers and women's center accounted for 85% and 88% of consolidated net revenues for the three month periods ended March 31, 1997 and 1996, respectively. Net revenues at the Company's acute care hospitals owned during both periods increased 7% for the three month period ended March 31, 1997 over the comparable prior year period while inpatient admissions at these facilities remained relatively unchanged. Patient days at the Company's acute care facilities owned during both periods decreased 2% during the 1997 first quarter as compared to the comparable prior year period due to a 2% decrease in the average length of stay. Outpatient activity at the Company's acute care hospitals continues to increase as gross outpatient revenues at the acute care facilities owned during both periods increased 11% for the three months ended March 31, 1997 as compared to the comparable prior year period and comprised 24% of the Company's acute care gross patient revenues during the first quarter of 1997 as compared to 23% during the 1996 first quarter. The increase is primarily the result of advances in medical technologies, which allow more services to be provided on an outpatient basis, and increased pressure from Medicare, Medicaid, health maintenance organizations (HMOs), preferred provider organizations (PPOs) and insurers to reduce hospital stays and provide services, where possible, on a less expensive outpatient basis. To accommodate the increased utilization of outpatient services, the Company has expanded or redesigned several of its outpatient facilities and services.

Page Eight of Twelve Pages

BEHAVIORAL HEALTH SERVICES

Net revenues from the Company's behavioral health services facilities accounted for 14% and 12% of the Company's consolidated net revenues for the three month periods ended March 31, 1997 and 1996, respectively. Net revenues at the Company's behavioral health centers owned during both periods decreased 2% for the three month period ended March 31, 1997 as compared to the comparable 1996 quarter. Admissions at the facilities owned during both periods increased 9% during the 1997 first quarter over the comparable 1996 quarter while patient days remained relatively unchanged due to 9% decrease in the average length of stay which decreased to 11.0 days in the 1997 quarter as compared to 12.1 days during the 1996 quarter. The reduction in the average length of stay is a result of changing practices in the delivery of behavioral health services and continued cost containment pressures from payors which includes a greater emphasis on the utilization of outpatient services. Management of the Company has responded to these trends by developing and marketing new outpatient treatment programs. The shift to outpatient care is reflected in higher revenues from outpatient services, as gross outpatient revenues at the Company's behavioral health services facilities owned during both periods increased 14% during the three months ended March 31, 1997 as compared to the comparable prior year period and comprised 18% of the Company's behavioral health services revenues for the three months ended March 31, 1997 as compared to 17% during the 1996 first quarter.

OTHER OPERATING RESULTS

Depreciation and amortization expense increased 29% or \$4 million for the three months ended March 31, 1997 as compared to the comparable prior year period due primarily to the 1996 acquisitions mentioned above.

Interest expense increased 7% or \$300,000 during the 1997 first quarter as compared to the comparable 1996 period due to the increased borrowings used to partially finance the 1996 acquisitions mentioned above. In June 1996, the Company issued four million shares of its Class B Common Stock at a price of \$26 per share. The total net proceeds of \$99.1 million generated from this stock issuance were also used to partially finance the 1996 purchase transactions mentioned above

The effective tax rate was 37% and 36% for the three month periods ended March 31, 1997 and 1996, respectively.

GENERAL TRENDS

An increased proportion of the Company's revenue is derived from fixed payment services, including Medicare and Medicaid which accounted for 51% and 48% of the Company's net patient revenues for the three months ended March 31, 1997 and 1996, respectively. The Company expects the Medicare and Medicaid revenues to continue to increase as a larger portion of the general population qualifies for coverage as a result of the aging of the population and expansion of state Medicaid programs. The Medicare program reimburses the Company's hospitals primarily based on established rates by a diagnosis related group for acute care hospitals and by cost based formula for behavioral health facilities.

In addition to the Medicare and Medicaid programs, other payors continue to actively negotiate the amounts they will pay for services performed. In general, the Company expects the percentage of its business from managed care programs, including HMOs and PPOs to grow. The consequent growth in managed care networks and the resulting impact of these networks on the operating results of the Company's facilities vary among the markets in which the Company operates.

Page Nine of Twelve Pages

In addition to the trends described above that continue to have an impact on operating results, there are a number of other more general factors affecting the Company's business. In May 1997, the President and members of Congress agreed on a tentative five year budget plan which calls for a \$115 billion reduction in the rate of increase in Medicare spending over the next five years. Included in this proposal are reductions in the future rate of increases to payments made to hospitals. The Company cannot predict whether the above proposal or any other proposal will ultimately be adopted, and if adopted, no assurance can be given that the implementation of such plans will not have a material adverse effect on the Company's business. In Texas, a law has been passed which mandates that the state senate apply for a waiver from current Medicaid regulations to allow the state to require that certain Medicaid participants be serviced through managed care providers. The Company is unable to predict whether Texas will be granted such a waiver or the effect on the Company's business of such waiver. Upon meeting certain conditions, and serving a disproportionately high share of Texas' and South Carolina's low income patients, three of the Company's facilities located in Texas and one in South Carolina became eligible and received additional reimbursement from each state's disproportionate share hospital fund. Included in the Company's financial results was an aggregate of \$8.1 million for the three months ended March 31, 1997 (including the additional reimbursement received by an acute care facility acquired during the second quarter of 1996) and \$1.8 million for the three months ended March 31, 1996 received pursuant to the terms of these programs. These programs are scheduled to terminate in the third quarter of 1997 and the Company cannot predict whether these programs will continue beyond their scheduled termination date.

LIQUIDITY AND CAPITAL RESOURCES

Net cash provided by operating activities was \$49 million for the three months ended March 31, 1997 and \$33 million for the three months ended March 31, 1996. The \$16 million increase during the 1997 three month period as compared to the 1996 comparable period was due primarily to a \$12 million increase in the net income plus the addback of the non-cash charges (depreciation, amortization and provision for self-insurance reserves) and \$4 million of other net working capital changes.

During the first quarter of 1997, the Company spent \$31 million to finance capital expenditures including a total of \$17 million on the construction of a new medical complex (including a 149-bed acute care facility) in Summerlin, Nevada and a new 130-bed replacement facility in Edinburg, Texas. The Company also reduced outstanding debt by \$17 million.

During the first quarter of 1997, the Company amended its commercial paper credit facility to increase the borrowing capacity to \$75 million from \$50 million and to reduce the commitment fee. As of March 31, 1997, the Company had \$206 million of unused borrowing capacity available under its commercial paper and revolving credit facilities.

Page Ten of Twelve Pages

PART II. OTHER INFORMATION

UNIVERSAL HEALTH SERVICES, INC. AND SUBSIDIARIES

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits:

- 10.1 Agreement of Limited Partnership of Direct Hospital Partners, L.P. (a District of Columbia Limited Partnership) by and among UHS of D.C., Inc. and The George Washington University.
- 10.2 Amendment No. 2, dated as of April 17, 1997 to Pooling Agreement dated as of November 16, 1993, among UHS Receivables Corp., a Delaware corporation, Sheffield Receivables Corporation, a Delaware corporation, and First Bank National Association, a national banking association, as trustee.
 - 27. Financial Data Schedule
- (b) Reports on Form 8-K

None

11. Statement re computation of per share earnings is set forth on Page six in Note 2 of the Notes to Condensed Consolidated Financial Statements.

All other items of this Report are inapplicable.

Page Eleven of Twelve Pages

UNIVERSAL HEALTH SERVICES, INC. AND SUBSIDIARIES

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Universal Health Services, Inc. (Registrant)

Date: May 12, 1997 /s/ Kirk E. Gorman

Kirk E. Gorman, Senior Vice President and Chief Financial Officer

(Principal Financial Officer and

Duly Authorized Officer).

Page Twelve of Twelve Pages

EXECUTION COPY

AGREEMENT OF

LIMITED PARTNERSHIP

0F

DISTRICT HOSPITAL PARTNERS, L.P. (a District of Columbia Limited Partnership)

by and among

UHS of D.C., Inc.

and

THE GEORGE WASHINGTON UNIVERSITY

Dated as of April 2, 1997

LIMITED PARTNERSHIP AGREEMENT

٥F

DISTRICT HOSPITAL PARTNERS, L.P.

This LIMITED PARTNERSHIP AGREEMENT (the "Agreement") is made and entered into as of April 2, 1997 (the "Effective Date"), by and among UHS of D.C., Inc., a Delaware corporation, as the general partner and a limited partner ("UHS"), and THE GEORGE WASHINGTON UNIVERSITY, a congressionally chartered institution in the District of Columbia, as a limited partner (the "University"). The parties to this Agreement shall also be referred to individually as a "Partner" and collectively as the "Partners."

RECITALS

WHEREAS, The George Washington University is a highly regarded university and academic medical center that has among its activities and affiliates The George Washington University Hospital (the "Hospital"), The George Washington University School of Medicine and Health Sciences (the "School of Medicine"), and The George Washington University faculty practice plan (the "Medical Faculty Associates").

WHEREAS, The George Washington University, through its School of Medicine and Medical Faculty Associates, trains physicians and allied health professionals through a program of classroom and clinical education that meets high academic standards.

WHEREAS, The Hospital is a 501-bed tertiary care hospital licensed as an acute care hospital in the District of Columbia, founded in 1944 and maintaining a strong commitment to community service.

WHEREAS, UHS is an affiliate of Universal Health Services, Inc. ("UHS Parent"), a nationally known health care organization that has expertise in managing and operating hospitals, developing regional networks, and providing efficient, cost-effective community health care.

WHEREAS, UHS and UHS Parent recognize the Hospital as an important asset of the District of Columbia community, recognize the academic mission and charitable purposes of the University and the School of Medicine, recognize the Hospital's significant contributions to the community, and have committed to provide community service.

WHEREAS, the Partners intend to bring their health care resources together in order to create efficiencies, compete effectively in the health care market, provide affordable, high quality health care services in the community, and enhance the graduate and undergraduate medical education programs of the School of Medicine as well as the Medical Faculty

Associates and attending physicians. An ongoing clinical and academic affiliation between the parties will meet their respective objectives by improving the training of health care professionals and the delivery of high quality health care services in the community.

WHEREAS, the Partners desire to form a limited partnership under the laws of the District of Columbia for the purposes and on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants contained herein, the Partners agree as follows:

- 1. Definitions. When used in this Agreement, the following terms shall have the meanings set forth below:
- 1.1. "Act" means the District of Columbia Uniform Limited Partnership Act of 1987, being District of Columbia Code Section 41-401 et seq., as amended.
- 1.2. "Adjusted Capital Account Deficit" means, with respect to any Partner, the deficit balance, if any, in such Partner's Capital Account as of the end of the relevant Fiscal Year, after giving effect to the following adjustments:
- (i) Credit to such Capital Account any amounts which such Partner is obligated to restore pursuant to any provision of this Agreement or is deemed to be obligated to restore pursuant to the penultimate sentences of Treasury Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5); and
- (ii) Debit to such Capital Account the items described in Sections 1.704- 1(b)(2)(ii)(d)(4), 1.704-1(b)(2)(ii)(d)(5), and 1.704-1(b)(2)(ii)(d)(6) of the Treasury Regulations.

The foregoing definition of "Adjusted Capital Account Deficit" is intended to comply with the provisions of Section 1.704-1(b)(2)(ii)(d) of the Treasury Regulations and shall be interpreted consistently therewith.

- 1.3. "Agreement" means this Agreement of Limited Partnership, as amended from time to time.
- 1.4. "Ancillary Agreements" means the Academic Affiliation Agreement, the HMO Hospital Services Agreement, the Guaranty Agreement, the Trademark License Agreement, the Parking Rental Agreement, the Management Agreement, and the Ground Lease which the Partners intend to execute as of the Transfer Date.

- 1.5. "Assignee" means a Person acquiring a Transferred Interest as set forth in Section 13.1.
- 1.6. "Bankruptcy" means the institution of any proceedings under federal or state laws for relief of debtors, including filing of a voluntary or involuntary petition in bankruptcy or the adjudication of a Person as insolvent or bankrupt, or the assignment of a Person's property for the benefit of creditors, or the appointment of a receiver, trustee or a conservator of any substantial portion of the Person's assets or the seizure by a sheriff, receiver, trustee or conservator of any substantial portion of the Person's assets, and the failure, in the case of any of these events, to obtain the dismissal of the proceeding or removal of the conservator, receiver or trustee within thirty (30) days of the event.
- 1.7. "Capital Account" means, for any given Partner, an account determined and maintained throughout the full term of the Partnership for such Partner in accordance with Treasury Regulation Section 1.704-1(b)(2)(iv). Subject to and in accordance with said Regulation, each Partner's Capital Account balance at any time shall equal such Partner's Capital Contribution increased by the Partner's allocable share of Partnership Profits, and decreased by Distributions made to the Partner by the Partnership and the Partner's allocable share of Partnership Losses. Upon the contribution to or Distribution from the Partnership of property, including money, in connection with the admission to or retirement from the Partnership of a Partner, respectively, the assets of the Partnership shall be revalued on the books of the Partnership to reflect the fair market value of such assets at the time of the occurrence of such event, and the Capital Accounts of the Partners shall be adjusted in the manner provided in Treasury Regulation Section 1.704-1(b) (2)(iv)(f).
- 1.8. "Capital Contribution" means, for any given Partner, the aggregate of the money and the fair market value of any property contributed by the Partner to the capital of the Partnership (net of any liability secured by such contributed property that the Partnership assumes or takes subject to as primary obligor) pursuant to Section 3 hereof.
 - 1.9. "Capital Plan" means the plan described in Section 3.7.2.
- 1.10. "Centers of Emphasis" means the six (6) clinical programs at the Hospital that the University and UHS have identified as areas of special emphasis and that are more fully described in Section 3.8.
- 1.11. "Chief Executive Officer" means an employee of either UHS or an affiliate of UHS to manage, supervise and administer the day-to-day operations of the Partnership.
- 1.12. "Code" means the Internal Revenue Code of 1986, as amended, or corresponding provisions of any successor statute.

- 1.13. "Depreciation" means, for each Fiscal Year, an amount equal to the depreciation, amortization, or other cost recovery deduction allowable for federal income tax purposes with respect to an asset for such Fiscal Year, except that if the Gross Asset Value of an asset differs from its adjusted basis for federal income tax purposes at the beginning of such Fiscal Year, Depreciation shall be an amount which bears the same ratio to such beginning Gross Asset Value as the federal income tax depreciation, amortization, or other cost recovery deduction for such Fiscal Year bears to such beginning adjusted tax basis; provided, however, that if the adjusted basis for federal income tax purposes of an asset at the beginning of such Fiscal Year is zero, Depreciation shall be determined with reference to such beginning Gross Asset Value using any reasonable method selected by the General Partner.
- 1.14. "Director" means a Person appointed to serve on the Partnership Board. $\ensuremath{\mathsf{Part}}$
- 1.15. "Distribution" means the aggregate of the money or the fair market value of any property distributed to Partners with respect to their Interests in the Partnership (net of any liability secured by such property that the Partner assumes or takes subject to as primary obligor), other than payments to Partners for services or as repayment of loans or advances.
 - 1.16. "Fiscal Year" means the period defined in Section 6.1.
- 1.17. "General Partner" means initially UHS and includes any successor, substitute or additional Person admitted to the Partnership as a General Partner as provided in Section 13 of this Agreement.
- 1.18. "Gross Asset Value" means, with respect to any asset, the asset's adjusted basis for federal income tax purposes, except the initial Gross Asset Value of any asset contributed by a Partner to the Partnership shall be the gross fair market value of such asset, as determined by the contributing Partner and the General Partner, and the Gross Asset Value of any asset distributed to any Partner shall be adjusted to equal its gross fair market value as of the date of distribution, as determined by the General Partner and the distributee partner, provided that the initial Gross Asset Values of the assets contributed to the Partnership pursuant to Sections 3.1(b) and 3.1(c) shall be as set forth in such Sections.
- 1.19. "Hospital" means the acute care hospital inpatient facility located at 901 23rd Street, N.W., Washington, D.C., or as renovated or relocated as described in Section 3.7.2.
- 1.20. "Initial Capital Loan" means the initial capital loan of UHS to the Partnership described in Section 3.1(c).
- 1.21. "Interest" means the entire ownership interest of a Partner in the Partnership at any particular time, including the right of such Partner to any and all benefits to

which a Partner may be entitled as provided in this Agreement, subject to the obligations of such Partner to comply with all of the terms and conditions of this Agreement.

- 1.22. "Letter of Credit" means the irrevocable bank letter of credit in the amount of Forty Million Dollars (\$40,000,000) described in Section 3.9, as the same may be renewed or modified.
- 1.23. "Limited Partner" means initially UHS and the University, and includes any successor, substitute or additional Person admitted to the Partnership as a limited partner as provided in Section 13 of this Agreement.
- 1.24. "Major Decisions" means any of the actions of the Partnership requiring approval of a majority of the Directors appointed by UHS and a majority of the Directors appointed by the University, as set forth in Section 7.3.
- 1.25. "Management Agreement" means the agreement between UHS or an affiliate of UHS and the Partnership for the provision of management services to the Hospital which the Partners intend to execute as of the Transfer Date.
- 1.26. "Nonrecourse Deductions" has the meaning set forth in Section 1.704- 2(b)(1) and 1.704-2(c) of the Treasury Regulations.
- 1.27. "Nonrecourse Liability" has the meaning set forth in Section 1.704- 2(b)(3) of the Treasury Regulations.
- 1.28. "Partner" means any General Partner or Limited Partner of the Partnership.
- 1.29. "Partner Nonrecourse Debt" has the same meaning as the term "partner nonrecourse debt" set forth in Section 1.704-2(b)(4) of the Treasury Regulations.
- 1.30. "Partner Nonrecourse Debt Minimum Gain" means an amount, with respect to each Partner Nonrecourse Debt, equal to the Partnership Minimum Gain that would result if such Partner Nonrecourse Debt were treated as a Nonrecourse Liability, determined in accordance with Section 1.704-2(i)(3) of the Treasury Regulations.
- 1.31. "Partner Nonrecourse Deductions" has the same meaning as the term "partner nonrecourse deductions" set forth in Sections 1.704-2(i)(1) and 1.704-2(i)(2) of the Treasury Regulations.
- 1.32. "Partnership" means District Hospital Partners, L.P., the limited partnership organized pursuant to this Agreement.

- 1.33. "Partnership Board" means the board of directors of the Partnership which board shall serve in an advisory capacity to the Partnership and the Partners, except for those Major Decisions requiring a vote of the Partnership Board under Section 7.3 hereof.
- 1.34. "Partnership Minimum Gain" has the meaning set forth in Sections 1.704-2(b)(2) and 1.704-2(d) of the Treasury Regulations.
- 1.35. "Percentage Interest" means, for purposes of allocating all items of Profits and Losses among the Partners, a ratio, expressed in percentages, reflecting a Partner's Interest in the Partnership relative to the Interests of other Partners. The initial Percentage Interests of the Partners are as set forth on Exhibit 1.35 hereto.
- 1.36. "Person" means any natural person, partnership, corporation, association, trust, governmental agency or other legal entity.
- 1.37. "Profits" and "Losses" means, for each Fiscal Year or other period, the Partnership's taxable income or loss for such year or period, determined in accordance with Section 703(a) of the Code (including in such taxable income or loss all items of income, gain, loss or deduction required to be stated separately), increased by any income of the Partnership that is exempt from federal income tax, and decreased by expenditures of the Partnership described in Section 705(a)(2)(B) of the Code or treated as Section 705(a)(2)(B)expenditures pursuant to Section 1.704-1(b)(2)(iv) of the Treasury Regulations. Profits and Losses, depreciation, depletion, amortization, gain, and loss with respect to any property (including intangibles) that under Sections 1.704-1(b)(2)(iv)(d) and (b)(2)(iv)(f) of the Treasury Regulations is properly reflected on the books of the Partnership at a book value that differs from the adjusted tax basis of such property shall be determined based on the book value of such property, in accordance with the principles of Section 1.704-1(b)(2)(iv)(g) of the Treasury Regulations. Any items which are specially allocated pursuant to the provisions of Sections 5.4 and 5.5 shall not be taken into account in computing Profits and Losses.
- 1.38. "Project Fund" means the account established for the purposes set forth in Section 3.7.2.
- 1.39. "Purposes" means the business and purposes of the Partnership as described in Section 2.4.
- 1.40. "Service Area" of the Partnership means the area described in Exhibit 1.40 to be attached hereto.
- 1.41. "Substitute Partner" means an Assignee admitted as a Partner as set forth in Section 13.1.

- 1.42. "UHS Contribution" means the initial Capital Contribution of UHS as set forth in Section 3.1(c).
- 1.43. "Transfer" means any voluntary or involuntary sale, assignment, transfer, exchange, lease, mortgage, charge, hypothecation, pledge or other conveyance or encumbrance, including any transfer by operation of law or otherwise.
- 1.44. "Transfer Date" means the Closing Date set forth in the Contribution Agreement between the University and the Partnership.
- 1.45. "Transferred Interest" means an Interest that is subject to a Transfer as set forth in Section 13.1.
- 1.46. "Treasury Regulation" means those regulations promulgated pursuant to the Code.
- 1.47. "University Contribution" means the University's initial Capital Contribution as set forth in Section 3.1(b).
- 1.48. "University Contribution Agreement" means the Contribution Agreement between the University and the Partnership to be executed as of the Transfer Date.
 - 2. Organization of Partnership.

2.1. Formation.

- (a) The Partners hereby form the Partnership pursuant to the provisions of the Act, which shall govern the relationship of the Partners except as expressly provided to the contrary herein;
- (b) The Partners shall take any and all actions as may from time to time be required under the laws of the District of Columbia to give effect to the Partnership and to maintain it in good standing; and
- (c) Except as provided herein, the Partnership shall be managed by the General Partner of the Partnership, having the powers specified in Section 7.1 hereof and subject to the limitations set forth in this Agreement.
- $\,$ 2.2. Name. The name of the Partnership shall be District Hospital Partners, L.P.
- 2.3. Principal Place of Business. The principal place of business of the Partnership shall be at the Hospital, which as of the date hereof is located at 901 23rd Street,

N.W., Washington, D.C. The Partnership shall continuously maintain a principal office in the District of Columbia, which office shall be located at its principal place of business. The Partnership may also have such other offices or places of business as the Partners may from time to time determine.

- 2.4. Purposes. The Purposes of the Partnership shall be (a) to own and operate an acute care hospital in the District of Columbia known as The George Washington University Hospital, (b) to provide or arrange for the provision of health care services or related services, (c) to furnish uncompensated and charity health care, (d) to develop and maintain an integrated delivery system, (e) to develop Centers of Emphasis, (f) to support the academic programs of the School of Medicine and the Medical Faculty Associates including research and teaching and (g) to engage in any other business and do any and all other acts and things agreed upon by the Partners and consistent with the foregoing.
- 2.5. Term. The term of the Partnership shall commence as of the Effective Date and shall continue for fifty (50) years (the "Term"), unless dissolved, liquidated and terminated sooner pursuant to the provisions of Section 14 hereof. At the end of the fifty (50) year term, UHS shall have right to extend the Term for an additional five (5) year renewal term and four (4) separate additional five (5) year renewal terms thereafter, provided that UHS provides the University with eighteen (18) months written notice prior to the end of the Term and each applicable renewal term thereafter.
- 2.6. Compliance With Securities Laws. Each Partner represents and warrants that it has acquired or is acquiring its Interest for its own account, and not with a view toward the resale thereof. Each of the parties hereto is fully aware and acknowledges that the offer and sale of the Interest which it has acquired or is acquiring has not been registered under the Securities Act of 1933, as amended, nor registered or qualified under the securities laws of any state or the District of Columbia. Each Partner acknowledges and agrees that its Interest may not be sold, transferred, pledged, or hypothecated without registration under such acts or an opinion of legal counsel that such transfer may be legally effected without such registration. Additional restrictions on Transfers of Interests are set forth in this Agreement.
- 2.7. Resident Agent. The name and address of the Partnership's resident agent in the District of Columbia shall be C T Corporation System, 1025 Vermont Avenue, N.W., Washington, D.C. 20005, until changed by the General Partner.

- 3. Capitalization of Partnership.
 - 3.1. Description, Timing and Value of Initial Capital Contributions.
- (a) Provided the conditions set forth in Subsection (e) below are satisfied, each Partner shall make its initial Capital Contribution to the Partnership on the Transfer Date unless otherwise mutually extended by the Partners.
- (b) The University shall transfer and assign to the Partnership, as its initial Capital Contribution (the "University Contribution"), all of its rights, title and interest in and to the Assets as defined in the University Contribution Agreement, subject to the obligations and liabilities described in the University Contribution Agreement.
- (c) UHS shall provide to the Partnership's Project Fund, as its initial capital commitment, Eighty Million Dollars (\$80,000,00). Seventy-two Million Dollars (\$72,000,000) of such initial capital commitment shall be in the form of a Capital Contribution (the"UHS Contribution") and the remaining eight million dollars (\$8,000,000) shall be in the form of an initial capital loan (the "Initial Capital Loan"). The UHS Contribution shall be allocated between UHS's Interests as General Partner and as a Limited Partner in accordance with the Percentage Interests of UHS set forth in Exhibit 1.35. Forty Million Dollars (\$40,000,000) shall be delivered by wire transfer of immediately available funds to the Partnership's account as of the Transfer Date. The remaining Forty Million Dollars (\$40,000,000) of UHS' capital commitment will be provided from time to time as required to fund capital expenditures in accordance with the Capital Plan and shall be secured by the Letter of Credit. The UHS Contribution and Initial Capital Loan shall be used only for the Project Fund.
- (d) UHS shall enter into a loan agreement with the Partnership whereby UHS shall make the Initial Capital Loan. The Initial Capital Loan amount shall be Eight Million Dollars (\$8,000,000) and shall be added to the Project Fund. The Initial Capital Loan shall be on a long term basis and shall bear interest at a rate of twelve percent (12%) per annum to be paid by the Partnership only to the extent that the Partnership has Profits.
- (e) On or before the Transfer Date, the Partners shall take the following actions and/or ensure that the following events occur and such actions and events shall be conditions precedent to either Partner's obligation to contribute its initial Capital Contribution:
- (i) The University and the Partnership shall have executed and performed all of the terms, covenants, representations and warranties set forth in the University Contribution Agreement, as well as the terms, covenants, representations and warranties set forth in this Agreement and the exhibits and schedules attached hereto.
- (ii) All parties shall have executed and delivered the Ancillary Agreements.

- $\mbox{(iii)}$ UHS shall have delivered to the Partnership the Loan Agreement.
- (iv) The Partnership shall have executed and/or delivered or caused to be delivered to each Partner the following:
- a. A certificate of the General Partner certifying that all consents and approvals that are required from any person, entity, governmental body or regulatory agency in connection with the consummation of the transactions contemplated by this Agreement by the Partnership have been obtained;
- b. A certificate of the corporate secretary of the General Partner certifying (i) the incumbency of the officers of the General Partner from the Effective Date to the Transfer Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement; and (ii) the minutes of the meeting of the shareholders of UHS and the resolutions of the Board of Directors of the General Partner authorizing the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, including the transfer of the UHS Contribution, and that such minutes and resolutions have not been amended or rescinded and remain in full force and effect;
- c. A favorable original certificate of good standing of the Partnership issued by the District of Columbia Department of Consumer and Regulatory Affairs; and
- d. Such other instruments, certificates, consents or other documents as may be reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof.
- (v) UHS shall have executed and/or delivered or caused to be delivered to the Partnership the following:
- a. A favorable original certificate of good standing issued by the Delaware Secretary of State;
- b. A certificate of the President or any Vice President of UHS certifying to the Partnership (i) the accuracy of the representations and warranties set forth in Section 4 hereof and (ii) that all consents and approvals that are required from any person, entity, governmental body or regulatory agency in connection with (a) the transfer of the UHS Contribution to the Partnership's Project Fund and (b) the execution, delivery and performance of the Management Agreement;
- c. A certificate of the corporate Secretary of UHS certifying to the Partnership (i) the incumbency of the officers of UHS from the Effective Date to the Transfer Date and bearing the authentic signatures of all such officers who shall execute this Agreement

and any additional documents contemplated by this Agreement; (ii) as to the resolutions of the Board of Directors of UHS authorizing (a) the transfer of the UHS Contribution by UHS to the Partnership, (b) the execution, delivery and performance of this Agreement and the Ancillary Agreements by UHS, and (c) the execution, delivery and performance of the Ancillary Agreements; (iii) that such resolutions have not been amended or rescinded and remain in full force and effect; and (iv) the current Articles of Incorporation and Bylaws of UHS;

- d. Copies of the documents executed and/or delivered in connection with the transfer of the UHS Contribution by UHS to the Partnership which are identified in Exhibit 3.1(d)(v)d attached hereto, and
- e. Such other certificates, consents, endorsements, assignments, assumptions, evidences, and other documents or instruments as may be reasonably requested by the Partnership in order to transfer the UHS Contribution to the Partnership's Project Fund, to carry out the transaction contemplated by this Agreement and to comply with the terms hereof.
- (vi) The University shall have executed and/or delivered or caused to be delivered to the Partnership and UHS the following:
- a. A certificate of the corporate secretary of the University certifying (i) the incumbency of the officers of the University from the Effective Date to the Transfer Date and bearing the authentic signatures of all such officers who shall execute this Agreement and any additional documents contemplated by this Agreement; (ii) the minutes of the meeting of the Trustees of the University and the resolutions of the Board of Trustees of the University authorizing the execution and delivery of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby, including the transfer of the University Contribution; (iii) that such minutes and resolutions have not been amended or rescinded and remain in full force and effect; and (iv) the current Charter and Bylaws of the University;
- b. A favorable original certificate of good standing of the University issued by the District of Columbia Department of Consumer and Regulatory Affairs or an applicable alternative certificate;
- c. Such other instruments, certificates, consents or other documents as may be reasonably necessary to carry out the transactions contemplated by this Agreement and to comply with the terms hereof; and
- d. Copies of the documents executed and/or delivered in connection with the transfer of the University Contribution to the Partnership, which are identified in the University Contribution Agreement.

- (vii) On the Transfer Date, no action or proceeding shall be pending or threatened wherein an unfavorable judgment, decree or order would, in the reasonable opinion of legal counsel for either Partner, prevent or make materially unfavorable the carrying out of this Agreement, or would cause the transactions contemplated by this Agreement to be rescinded. In the event of the receipt of any communication from any department or agency of government or any other notice (a copy of which communication or notice shall be promptly delivered to the Partnership and each Partner) prior to the closing with regard to the transactions contemplated by this Agreement, which communication or notice shall, in the reasonable opinion of legal counsel, threaten such an action or proceeding, either Partner may terminate this Agreement by giving thirty (30) days advance written notice to the other Partner. During such thirty (30) day period, UHS and the University shall meet and confer in good faith to attempt to resolve the issue which is the subject of the action or proceeding. In the event the parties are unable to do so, or fail to agree to extend the thirty (30) day period, either Partner may terminate this Agreement by giving written notice to the other Partner and all parties shall thereupon be released from any and all liability related to this Agreement.
- (f) In the event that each of the conditions precedent set forth in Subsection (e) above shall not have been satisfied by the Transfer Date, the Partners shall not be required to make the University Contribution or the UHS Contribution and the Partnership shall be dissolved, pursuant to the provisions of Section 14.
- 3.2. Additional Limited Partners. Subject to the provisions of Section 7.3 and Section 13, the Partnership may from time to time elect to offer limited partnership interests to other Persons, as permitted by law. Any Limited Partner admitted to the Partnership after the execution of this Agreement shall make an initial Capital Contribution in an amount and at a time as determined by the Partnership in exchange for its Limited Partnership Interest. A new Limited Partner's initial Capital Contribution under this Section 3.2 shall, unless otherwise determined by the Partnership, equal the following amount:
 - (i) the Percentage Interest of the new Partner, multiplied by
- (ii) the fair market value of all the Partnership's assets (including any new Limited Partner's initial Capital Contribution) as of the effective date of admission.
- 3.3. Additional Capital Requirements. It shall be the general policy of the Partnership that all monies necessary to carry on the activities of the Partnership shall be obtained first through funds derived from the operation of the Partnership. To the extent funds derived from the Partnership are not sufficient, it is the intent of the Partners that the Partners will contribute an additional Forty-Five Million Dollars (\$45,000,000) to the Partnership in excess of the University Contribution, the UHS Contribution and the Initial Capital Loan for Hospital operations over a ten (10) year period following the Transfer Date. Additional capital including such Forty-five Million Dollars (\$45,000,000) will, at the election of the General Partner, be, in whole or in part, in the form of additional Capital Contributions

or capital loans as described in Section 3.4 hereof or loans from non-Partners. No Partner shall be required to provide additional capital to the Partnership for the Capital Plan until the Project Fund is fully depleted.

- 3.4. Capital Contributions and Capital Loans. Partners may be required to make additional Capital Contributions to the Partnership in the amounts and at the times determined by the General Partner. No Partner may make a voluntary Capital Contribution without the consent of the General Partner. All additional Capital Contributions shall be made in accordance with each Partner's Percentage Interest in the Partnership, unless the Partners agree otherwise. In the event a Partner fails to make an additional Capital Contribution, the General Partner shall make the Capital Contribution in accordance with Sections 3.4 and 3.6. The Partnership may borrow funds from any of the Partners on reasonable terms and conditions approved by the Partners at the time such loan or advance is obtained, subject to Section 7.3. Such loans or advances shall not increase or otherwise affect the Capital Accounts of any of the parties hereto. Funds loaned by a Partner to the Partnership for purposes of satisfying the need for an additional Capital Contribution shall accrue interest at the prime rate as determined from time to time (as published in the Wall Street Journal) plus two percent (2%) per annum and shall be repaid prior to any Distribution.
- 3.5. No Withdrawals from Capital Accounts. No Partner shall be entitled to withdraw funds from its Capital Account, except pursuant to a Distribution made in accordance with Section 5 or the liquidation of the Partnership in accordance with Section 14, unless the Partners consent thereto.
- 3.6. Failure to Make Capital Contributions. The Interest of a Partner who fails to make any required Capital Contribution or other payment to the Partnership (subsequent to its initial Capital Contribution) shall be reduced upon failure to make such Capital Contribution or payment, and each Partner's Percentage Interest thereafter shall be recalculated.

3.7. Project Fund.

- 3.7.1. On the Transfer Date, the Partnership shall establish the Project Fund, which the Partnership shall use to implement the Capital Plan described in Section 3.7.2. The Project Fund shall consist of the proceeds of the UHS Contribution and of the Loan Agreement and any additional Capital Contributions as determined by the General Partner pursuant to Section 3.3 together with income on such funds.
- 3.7.2. The parties to this Agreement shall work diligently to develop, approve, and implement a Capital Plan to be prepared by the General Partner to substantially reconstruct, renovate and equip the existing Hospital facility, using the Project Fund. Only after the Capital Plan has been fully implemented and construction is complete, any funds remaining in the Project Fund shall be used to develop an integrated delivery system

including, without limitation, linkage with the Medical Faculty Associates (as that term is defined in the Academic Affiliation Agreement attached hereto as Exhibit 3.7.2), development of a regional healthcare delivery infrastructure (including expansion of an ambulatory care network) and continued development of a primary care physician network.

- 3.7.3. The Project Fund shall not be used to pay management fees under the Management Agreement, offset deficits in budget pools under the HMO Hospital Service Agreement, make academic support payments under the Academic Affiliation Agreement, or in any way fund operating deficits of the Partnership.
- 3.8. Centers of Emphasis. The parties to this Agreement intend that within two (2) years following the Transfer Date, the General Partner shall develop strategic, capital and operating plans for each of the Centers of Emphasis to provide capital, marketing and faculty support for the Centers of Emphasis, each of which shall be subject to the approval of the University under Section 7.3 hereof. "Centers of Emphasis" are the following six (6) clinical programs at the Hospital that have been identified by the University as areas of special emphasis: (1) cardiovascular disease, (2) cancer treatment and research, (3) emergency medicine, (4) minimally invasive surgery, (5) neurologic disease and (6) women's health.
- 3.9. Letter of Credit. The Letter of Credit shall be issued in favor of the Partnership, effective as of the Transfer Date. The Letter of Credit may be drawn by the Partnership in the event UHS fails to contribute to the Partnership the deferred portion of the UHS Contribution or fails to make available the Initial Capital Loan as the same are required from time to time to fund continuing capital expenditures in accordance with the Capital Plan. From and after such time as UHS contributes any of the deferred portion of the UHS Contribution and makes any portion or all of the Initial Capital Loan to the Partnership, UHS shall have the right to reduce the stated amount of the Letter of Credit such that the sum of contributions actually made to the Project Fund from the UHS Contribution and the Initial Capital Loan and the stated amount of the Letter of Credit equals Eighty Million Dollars (\$80,000,000). The Letter of Credit shall be in a form, and issued by a financial institution, reasonably acceptable to the University. The University will review in good faith the need for the continued existence of the Letter of Credit to secure the ongoing obligation of UHS to complete the UHS Contribution, taking into consideration the progress of construction under the Capital Plan, the credit rating of UHS, and other factors.

UHS shall bear the cost of the Letter of Credit, except that the University shall reimburse UHS on an annual basis for its cost of the Letter of Credit plus fifty (50) basis points up to a maximum of 1.75% of the stated amount of the Letter of Credit not to exceed eight million dollars (\$8,000,000) stated amount; provided that for each dollar of reduction in the amount of the face amount of the Letter of Credit other than as a result of the cash funding of UHS Contribution, the amount of the Letter of Credit with respect to which the University is obligated to pay UHS its cost shall be reduced correspondingly.

- 3.10 Initial Capital Loan. Only after UHS contributes all of the deferred portion of the UHS Contribution, the Initial Capital Loan may be drawn upon from time to time to fund continuing capital expenditures in accordance with the Capital Plan or after completion of the Capital Plan otherwise in accordance with Section 3.7.2.
- 4. Representations and Warranties of UHS. UHS hereby represents and warrants to the Partnership, which representations and warranties shall be true and correct on the Effective Date, as follows:
- 4.1. Organization; Good Standing. UHS is a Delaware corporation, duly organized, validly existing and in good standing under the laws of Delaware with full corporate power and authority to carry on its businesses. UHS is also duly qualified to carry on its businesses in the District of Columbia.
- 4.2. Authority; Validity; No Breach. UHS has the full right, power, legal capacity and authority, to execute, deliver and carry out the terms of this Agreement and all documents and agreements necessary to give effect to the provisions of this Agreement and to consummate the transactions contemplated hereby. All corporate and other actions required to be taken by UHS to authorize the execution, delivery and performance of this Agreement, all documents executed by it necessary to give effect to this Agreement, and all transactions contemplated hereby have been duly and properly taken or obtained or will be duly and properly taken or obtained by UHS prior to the Transfer Date. No other corporate or other action on the part of UHS is necessary to authorize the execution, delivery and performance of this Agreement, all documents necessary to give effect to this Agreement and all transactions contemplated hereby. Any consent which has not been obtained would not have an adverse effect on the transactions contemplated hereby.

This Agreement is, and the documents to be delivered at closing will be, the lawful, valid and legally binding obligations of UHS in accordance with their respective terms. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby will not, with or without the giving of notice and/or the passage of time: (a) violate the Articles of Incorporation or Bylaws of UHS, or any provision of law, statute, rule or regulation to which UHS is subject; or (b) violate or conflict with any judgment, order, writ or decree of any court applicable to UHS; which violation or conflict would have a material adverse effect on the transactions contemplated hereby.

- 4.3. Consents and Approvals. UHS has obtained each consent, approval, permit, waiver, authorization or other action of or by any court, governmental or nongovernmental person or entity, that is required in connection with the execution, delivery or performance of this Agreement by UHS.
- 4.4. Solvency. UHS is not insolvent and will not be rendered insolvent as a result of any of the transactions contemplated by this Agreement. For purposes hereof, the

term "solvency" means that: (a) the fair salable value of UHS's tangible assets is in excess of the total amount of its liabilities (including for purposes of this definition all liabilities, whether or not reflected on a balance sheet prepared in accordance with generally accepted accounting principles, and whether direct or indirect, fixed or contingent, secured or unsecured, and disputed or undisputed); (b) UHS is able to pay its debts or obligations in the ordinary course as they mature; and (c) UHS has capital sufficient to carry on its businesses and all businesses in which it is about to engage.

- 4.5. Guaranty. UHS Parent will execute a guaranty agreement with UHS (the "Guaranty Agreement"), a copy of which is attached hereto as Exhibit 4.5.
- 4.6. Community Service. UHS recognizes the significant contribution the University has made to the local community through its support of uncompensated and indigent care at the Hospital. For at least two (2) years from the Transfer Date, the Partnership shall maintain the Hospitals policies for the treatment of indigent patients as such policies exist as of the Transfer Date. Moreover, UHS recognizes as a member of the community the Partnership's responsibilities for the treatment of indigent patients. Notwithstanding anything in this Agreement or any Ancillary Agreement to the contrary, the Partnership shall comply with all applicable laws and regulations relating to uncompensated care and community service, including but not limited to the requirements of any applicable Certificate of Need law.

5. Distributions and Allocations.

- 5.1. Distributions. Subject to Section 14 below, the General Partner may make Distributions to the Partners of cash or other property of the Partnership. No Distribution under this Section 5.1 shall be made, however, unless the cash or property to be distributed is determined by the General Partner, in accordance with sound business practices, to be in excess of the Partnership's business needs, including for this purpose any and all reasonable reserves for working capital or liabilities of the Partnership, such as the Partnership's repayment obligations with respect to any Capital Loans, or other reasonably foreseeable contingencies. All Distributions shall be in proportion to the Partners' respective Percentage Interests.
- 5.2. Record Date for Distributions. Each Distribution to the Partners pursuant to this Agreement shall be made to the Partners who are holders of record of Interests as of the date such Distribution is approved.
- 5.3. Allocations of Profits and Losses. Except as otherwise provided in Sections 5.4 and 5.5, Profits and Losses shall be allocated among the Partners annually at the close of each Fiscal Year of the Partnership according to their respective Percentage Interests.

5.4. Special Allocations. The following special allocations shall be made in the following order:

5.4.1. Minimum Gain Chargeback. Except as otherwise provided in Section 1.704-2(f) of the Treasury Regulations, notwithstanding any other provision of this Section 5, if there is a net decrease in Partnership Minimum Gain during any Fiscal Year, each Partner shall be specially allocated items of Partnership income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Partner's share of the net decrease in Partnership Minimum Gain, determined in accordance with Treasury Regulations Section 1.704-2(g). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Partner pursuant thereto. The items to be so allocated shall be determined in accordance with Sections 1.704-2(f)(6) and 1.704-2(j)(2) of the Treasury Regulations. This Section 5.4.1 is intended to comply with the minimum gain chargeback requirement in Section 1.704-2(f) of the Treasury Regulations and shall be interpreted consistently therewith.

5.4.2. Partner Minimum Gain Chargeback. Except as otherwise provided in Section 1.704-2(i)(4) of the Treasury Regulations, notwithstanding any other provision of this Section 5, if there is a net decrease in Partner Nonrecourse Debt Minimum Gain attributable to a Partner Nonrecourse Debt during any Fiscal Year, each Partner who has a share of the Partner Nonrecourse Debt Minimum Gain attributable to such Partner Nonrecourse Debt, determined in accordance with Section 1.704-2(i)(5) of the Treasury Regulations, shall be specially allocated items of Partnership income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Partner's share of the net decrease in Partner Nonrecourse Debt Minimum Gain attributable to such Partner Nonrecourse Debt, determined in accordance with Treasury Regulations Section 1.704-2(i)(4). Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Partner pursuant thereto. The items to be so allocated shall be determined in accordance with Sections 1.704-2(i)(4) and 1.704-2(j)(2) of the Treasury Regulations. This Section 5.4.2 is intended to comply with the minimum gain chargeback requirement in Section 1.704-2(i)(4) of the Treasury Regulations and shall be interpreted consistently therewith.

5.4.3. Qualified Income Offset. In the event any Partner unexpectedly receives any adjustments, allocations, or distributions described in Section 1.704-1(b)(2)(ii)(d)(4), Section 1.704-1(b)(2)(ii)(d)(5), or Section 1.704-1(b)(2)(ii)(d)(6) of the Treasury Regulations, items of Partnership income and gain shall be specially allocated to each such Partner in an amount and manner sufficient to eliminate, to the extent required by the Treasury Regulations, the Adjusted Capital Account Deficit of such Partner as quickly as possible, provided that an allocation pursuant to this Section 5.4.3 shall be made only if and to the extent that such Partner would have an Adjusted Capital Account Deficit after all other allocations provided for in this Section 5 have been tentatively made as if this Section 5.4.3 were not in the Agreement.

- 5.4.4. Gross Income Allocation. In the event any Partner has a deficit Capital Account at the end of any Fiscal Year which is in excess of the sum of (i) the amount such Partner is obligated to restore pursuant to any provision of this Agreement, and (ii) the amount such Partner is deemed to be obligated to restore pursuant to the penultimate sentences of Sections 1.704-2(g)(1) and 1.704-2(i)(5) of the Treasury Regulations, each such Partner shall be specially allocated items of Partnership income and gain in the amount of such excess as quickly as possible, provided that an allocation pursuant to this Section 5.4.4 shall be made only if and to the extent that such Partner would have a deficit Capital Account in excess of such sum after all other allocations provided for in this Section 5 have been made as if this Section 5.4.4 were not in the Agreement.
- 5.4.5. Partner Nonrecourse Deductions. Any Partner Nonrecourse Deductions for any Fiscal Year shall be specially allocated to the Partner who bears the economic risk of loss with respect to the Partner Nonrecourse Debt to which such Partner Nonrecourse Deductions are attributable in accordance with Treasury Regulations Section 1.704-2(i)(1).
- 5.4.6. Nonrecourse Deductions. Nonrecourse Deductions for any Fiscal Year shall be specially allocated among the Partners in proportion to their Percentage Interests.
- 5.5. Curative Allocations. The allocations set forth in Sections 5.4.1, 5.4.2, 5.4.3, 5.4.4, 5.4.5 and 5.4.6 (the "Regulatory Allocations") are intended to comply with certain requirements of the Treasury Regulations. It is the intent of the Partners that, to the extent possible, all Regulatory Allocations shall be offset either with other Regulatory Allocations or with special allocations of other items of Partnership income, gain, loss, or deduction pursuant to this Section 5.5. Therefore, notwithstanding any other provision of this Section 5 (other than the Regulatory Allocations), the General Partner shall make such offsetting special allocations of Partnership income, gain, loss, or deduction in whatever manner it determines appropriate so that, after such offsetting allocations are made, each Partner's Capital Account balance is, to the extent possible, equal to the Capital Account balance such Partner would have had if the Regulatory Allocations were not part of the Agreement. In exercising its discretion under this Section 5.5, the General Partner shall take into account future Regulatory Allocations under Section 5.4.1 and 5.4.2 that, although not yet made, are likely to offset other Regulatory Allocations previously made under Section 5.4.5 and 5.4.6.
- 5.6. Book Value/Tax Basis Differentials. In accordance with Section 704(b) and 704(c) of the Code and the Treasury Regulations promulgated thereunder, income, gain, loss, deductions and credit with respect to (i) any property contributed to the capital of the Partnership, and (ii) any property revalued on the Partnership's books in accordance with Treasury Regulation Section 1.704-1(b)(2)(iv)(f), shall, solely for tax purposes, be allocated among the Partners so as to take into account any variation between the adjusted basis of such

property to the Partnership for federal tax purposes and its value on the books of the Partnership. The General Partner shall make such allocations using the "remedial allocation method," as specified in Treasury Regulation Section 1.704-3(d).

5.7. Allocation in Respect of Interests Held for Partial Year. In the event of a change in the Partners' Percentage Interests during any Fiscal Year, Profits and Losses, if any, for such Fiscal Year shall be divided equally among the months of such year, and each Partner shall participate in the allocation for each month on the basis of its Percentage Interest as of the last day of that month. Notwithstanding the foregoing, the Partnership shall in all events utilize an accounting method for determining the allocable shares of the above items to which the Partners are entitled that complies with Code Section 706(d) and the Treasury Regulations thereunder.

6. Reports, Books and Records.

- 6.1. Maintenance. The Partnership shall maintain complete and accurate books and records, as required by the Act, at its principal office or such other place or places as the General Partner may from time to time determine. The Partnership shall adopt the same fiscal year as UHS uses in its business, or such other fiscal year as is required to be adopted by the Partnership for federal income tax purposes under Section 706 of the Code (a "Fiscal Year").
- 6.2. Financial Statements and Income Tax Returns. The General Partner shall cause to be prepared and delivered to each Partner, within one hundred twenty (120) days after the expiration of each Fiscal Year, a balance sheet and a profit and loss statement of the Partnership, a statement showing the Capital Accounts, the Distributions and the share in the Profits and Losses of each of the Partners for such Fiscal Year, and copies of all income tax informational returns filed by or on behalf of the Partnership, together with such additional tax information with respect to the Partnership's operations as shall be necessary for the preparation by the Partners of their federal or state income tax returns. Each of the Partners shall be entitled to request such additional reports on the operations or financial condition of the Partnership as they reasonably believe is appropriate.
- 6.3. Inspections. Each Partner of record shall have the right to examine, at any reasonable time or times for all purposes, the books and records of account, and the minutes and records of the Partnership, and to make copies thereof at such Partner's expense. Such inspection may be made by any agent or attorney of the Partner.
- 6.4. Tax Elections; Adjustment of Basis of Partnership. Upon the Transfer of any Interest in the Partnership in the manner provided in Section 743 of the Code, or a Distribution of property made in the manner provided in Section 734 of the Code, in the reasonable discretion of the General Partner, the Partnership may file an election under Section 754 of the Code to adjust the basis of the assets of the Partnership under the circumstances and

in the manner provided in Sections 734 and 743. In the event of such election, the General Partner shall take any and all necessary steps to consummate such adjustment, including, without limitation, the filing of the election with the income tax return of the Partnership for the first Fiscal Year to which the election applies.

The General Partner may (but need not), in its sole and absolute discretion, make any other elections under the Code or the Treasury Regulations, and the General Partner shall be absolved from all liability for any and all consequences to any previously admitted or subsequently admitted partners resulting from the making or failing to make any tax election.

6.5 Tax Matters Partner. The General Partner is hereby designated as the "tax matters partners" in accordance with Section 6231(a)(7) of the Code, and, in connection therewith and in addition to all other powers given thereto, shall have all powers necessary and appropriate to perform such role and to expend Partnership funds for professional services and costs associated therewith.

7. Management of the Partnership.

7.1. Responsibilities and Authority of the General Partner. The overall management and control of the Partnership and all of its affairs shall be in the General Partner, except as otherwise set forth herein or in the Management Agreement. The General Partner shall oversee the implementation of the decisions of the Partners and the day-to-day management of the Partnership by the Chief Executive Officer. The General Partner shall devote such attention and business capacity to the affairs of the Partnership as may be reasonably necessary to fully perform its duties as General Partner.

7.2. Partnership Board of Directors.

- 7.2.1. The board of directors of the Partnership (the "Partnership Board") shall consist of six (6) Directors. Three (3) Directors shall be appointed by the University and shall be University Directors, and three (3) Directors shall be appointed by UHS and shall be UHS Directors.
- 7.2.2. The University shall elect the initial Chairman of the Partnership Board after consultation with UHS. Thereafter, the Chairman shall be elected by the Partnership Board of Directors on the basis of the Partners' Partnership Interests.
- 7.2.3. The initial Directors shall be as set forth in Exhibit 7.2.3 attached hereto. The term of each of the initial Directors shall be from the inception of the Partnership until the first meeting of the Partners, at which time their successors shall be elected and qualified by the University and UHS, respectively. The University and UHS shall retain the discretion to replace any of their respective elected Directors during their term as Directors. After the first meeting of the Partners, all Directors shall serve for terms of one (1)

year, or until their successors shall have been elected and qualified or until the earlier of their death, removal, resignation or disqualification. After the first meeting, each Partner shall annually elect its class of Directors. Directors elected by a Partner shall be automatically disqualified as Directors in the event that the Partner ceases to be a Partner. There shall be no limitation on the number of terms a Director may serve.

- 7.2.4. In all events, a Director shall hold office until his successor has been elected and qualified.
- 7.2.5. An annual meeting of the Partnership Board shall be held at the date, time and place determined by the Partnership Board. Special meetings of the Partnership Board may be called by the Chairman, the Partnership Board or any two (2) Directors by giving at least five (5) days written notice to the Directors, except in the case of an emergency, which shall require at least forty-eight (48) hours written notice, of the date, time, place and purpose of the special meeting. Participation at any meeting may be by any means of communication pursuant to which all Directors participating may simultaneously hear each other.
- 7.2.6. Any Director may waive notice of any meeting. Except as set forth in the following sentence, the waiver must be in writing, signed by the Director entitled to the notice and filed with the minutes or other records of the Partnership. A Director's attendance at or participation in a meeting shall constitute a waiver of notice of that meeting unless the Director is attending for the sole purpose of objecting to the notice. Neither the business to be transacted at, nor the purpose of, any meeting of the Directors (other than for special meetings) need be specified in the notice or waiver of notice of the meeting.
- 7.2.7. The presence at a meeting of Directors representing a majority of the total voting power of the Partnership Board shall constitute a quorum for such meeting. If less than a quorum of the Directors is present at a meeting, the Directors present shall adjourn the meeting to a different time.
- 7.2.8. Except as otherwise provided in this Agreement, the Directors appointed by a particular Partner shall, as a group, have voting rights equivalent to that Partner's Percentage Interest. With the exception of the Major Decisions described in Section 7.3, the affirmative vote of a majority of the Percentage Interests represented at a meeting at which a quorum is present shall be required for an action of the Directors.
- 7.2.9. The Partner which elected a Director may remove the same Director with or without cause. A Director may resign at any time by giving written notice to the Directors or to the Chairman. A resignation is effective when the notice is given unless the notice specifies a future date. The pending vacancy may be filled before the effective date, but the successor shall not take office until the effective date.

- 7.2.10. A vacancy created by removal, death, incapacity, resignation or an increase in the number of Directors or by any other reason may be filled only by election by the Partner which elected such Director or, in the case of an increase in the number of Directors, by the Partner entitled to elect the new Directors. The individual elected to fill a vacancy shall serve until the next annual appointment of Directors by the Partners.
- 7.2.11. A Director who is present at a meeting of the Directors at which any official action is taken shall be conclusively presumed to have assented to the action unless he announces his dissent at the meeting, or his dissent is entered in the minutes of the meeting, or he files his written dissent with the secretary of the meeting before the meeting is adjourned.
- 7.2.12. Any action required or permitted to be taken at a meeting of the Directors may be taken without a meeting if a written consent setting forth the action so taken is signed by each Director and is filed with the minutes of proceedings of the Directors.
- 7.2.13. No Director shall receive a salary from the Partnership solely on account of his service as a Director, but this shall not preclude the Partnership from reimbursing Directors for expenses reasonably incurred in connection with their service or from paying Directors for other services rendered to the Partnership.
- 7.3. Major Decisions. While the University's Percentage Interest is twenty percent (20%) or greater or if less than twenty percent (20%), until the Capital Plan has been fully implemented (with respect to Section 7.3(h) only), the approval of a majority of the Directors appointed by the University and a majority of the Directors appointed by UHS shall be required for the following actions:
 - (a) Material amendments to this Agreement;
- (b) Sale or transfer of more than ten percent (10%) of the assets of the Partnership;
- (c) Addition of Partners to the Partnership or sale of any Partnership Interest, provided that approval shall not be unreasonably withheld;
- (d) Incurrence of debt by the Partnership which results in debt as a percentage of total capitalization in excess of fifty percent (50%);
- (e) Annual capital expenditures (not including those incurred as part of the Project Fund) in excess of fifteen percent (15%) of the net book value of the Partnership's total assets, with any unused portion to be carried over to future years;

- $% \left(1\right) =\left(1\right) \left(1\right) +\left(1\right) +\left(1\right) \left(1\right) +\left(1\right) \left(1\right) +\left(1\right) +\left(1\right) \left(1\right) +\left(1\right) +\left(1\right) \left(1\right) +\left(1$
- (g) Agreements between the Partnership and either the University or UHS or any affiliate of either party, unless such agreements are on terms and conditions at least as favorable as available from third parties;
- (h) Approval of the Project Fund and plans for the support and development of the Centers of Emphasis, and material modifications thereto; and
- (i) Acceptance of a business opportunity pursuant to Section 12.2.
- 7.4. Advisory Powers. In addition to the powers expressly set forth in this Agreement, the Partnership Board shall advise the Partnership on other matters affecting the Partnership. The Partnership Board shall receive advice from the Hospital Board of Trustees regarding Hospital matters.
- 8. Hospital Board of Trustees. The Partnership shall establish a board of twelve (12) trustees (the "Board of Trustees"). At least seven (7) of the trustees will be residents of the Hospital service area so as to provide local guidance to the policy and direction of the Hospital. The University and UHS shall each appoint six (6) Trustees, provided that no more than four (4) trustees shall be employed by or affiliated with the University and no more than four (4) trustees shall be employed by or affiliated with UHS. No less than four (4) trustees shall be members of the community who are not employed by or affiliated with UHS or the University. The responsibilities of the Board of Trustees shall be (a) establishing and maintaining accreditation and meeting accreditating agency requirements relating to medical staff credentialing, quality assurance, and oversight of Hospital responsibilities, (b) amending the Hospitals medical staff bylaws, rules and regulations, (c) promoting community involvement and community service and (d) advising the Partnership Board regarding Hospital matters. The Bylaws of the Board of Trustees are attached hereto as Exhibit 8.
- 9. Chief Executive Officer. The General Partner shall consult with the University with respect to the selection of the Chief Executive Officer of the Hospital. The Chief Executive Officer shall be appointed by, and shall serve at the pleasure of, The Partnership Board. The Chief Executive Officer shall be responsible for managing, supervising and administering the day-to-day operations of the Partnership.
- 10. Reserve Powers. Notwithstanding anything in this Agreement or any Ancillary Agreement to the contrary, and regardless of the Partnership Interest, if any, held by the University, the following actions of the Partnership shall require the approval of the University, which actions shall not affect the University's status as a Limited Partner or provide the University with any power to bind the Partnership:

- 10.1. Any material discontinuance, addition, or transfer outside of the Washington Circle area of the District of Columbia of any significant medical services provided at the Hospital as of the Transfer Date; provided that the University's approval shall not be unreasonably withheld. For purposes of this Section 10.1, the University's reasonable determination shall take into account such factors as:
- (a) Whether the discontinuance, addition, or transfer of services will materially adversely affect academic programs;
- (b) Whether the continuation of a service would impose a significant financial burden on the Partnership;
- (c) Whether the services to be discontinued or transferred are available in the community, subject to considerations of cost, quality, and reasonable access by patients and physicians; and
- (d) Whether the discontinuance, addition, or transfer of services is in the financial and strategic interest of the Partnership;
- 10.2. Any material discontinuance, addition, or transfer outside of the Washington Circle area of the District of Columbia of (a) any programs of undergraduate and graduate medical education sponsored or offered by The George Washington University at the Hospital as of the Transfer Date; (b) clinical programs conducted by the Medical Faculty Associates at the Hospital as of the Transfer Date; (c) Clinical Support Positions, as defined in the Academic Affiliation Agreement; and (d) Centers of Emphasis;
- 10.3. Selection of Clinical Chiefs as provided in the Academic Affiliation Agreement; and
- 10.4. Discontinuance of operation of an acute care hospital in the Washington Circle area of the District of Columbia.
 - 11. Confidential and Proprietary Information.
- 11.1. Confidential Information. Each Partner recognizes that due to the nature of this Agreement, it will have access to information of a proprietary nature owned by the Partnership, another Partner, and/or affiliates of another Partner including but not limited to documents and programs (whether or not completed or in use), operating manuals or similar materials that constitute nonmedical systems, policies and procedures, and methods of doing business, administrative, advertising or marketing techniques, financial affairs, and other proprietary information (collectively, the "Confidential Information"). Consequently, each Partner acknowledges and agrees that the Partnership, the other Partner(s) and the affiliates of the other Partner(s) respectively, have a proprietary interest in such Confidential Information

and that all such information constitutes confidential and proprietary information and/or trade secret property of the respective party or parties. Each Partner hereby expressly and knowingly waives any and all right, title and interest in and to other parties' Confidential Information and agrees to return all copies of Confidential Information to the provider(s) of such information at the Partner's expense upon the expiration or earlier termination of this Agreement; provided, however, that any Confidential Information developed and owned by the Partnership shall continue to be available for use by the Partners and their affiliates following the expiration or termination of this Agreement. Notwithstanding the foregoing, any Confidential Information used by the Partners after the expiration or termination of this Agreement shall remain confidential and proprietary and shall not be shown or disclosed to Persons other than the Partner's affiliates without the prior written consent of the party or parties furnishing the Confidential Information or except: (1) as required in governmental filings or judicial, administrative or arbitration proceedings, or (2) as otherwise required by law (including but not limited to in response to a subpoena from a court or authorized governmental agency) (hereinafter the "Confidentiality Exception"). All provider cost and rate information and other comparable data deemed by a party to be proprietary shall only be furnished to the other party or parties and their representatives (or, if deemed by counsel to be appropriate in connection with compliance with applicable antitrust or similar laws, to a neutral party), who shall not release it to any other Person without a consent of the party or parties owning such data or unless a Confidentiality Exception applies.

Each Partner further acknowledges and agrees that the Partnership, other Partner(s), and other Partner(s)' affiliates, respectively, are entitled to prevent their respective competitors from obtaining and utilizing the Confidential Information. Therefore, each Partner agrees to hold the Confidential Information in strictest confidence and to not disclose such information or allow such information to be disclosed, directly or indirectly, during and after the Term of this Agreement to any Person or entity other than those Persons or entities who are legal and financial advisors or employees of or are otherwise affiliated with the Partnership, a Partner or an affiliate of a Partner, or the Partnership's affiliates (collectively, the "Partnership Representatives") on a need to know basis, without the prior written consent of the party or parties furnishing Confidential Information unless a Confidentiality Exception applies. Further, each Partner shall require its Partnership Representatives that receive the Confidential Information to abide by the terms of this Section 11.1. Neither the Partnership, a Partner nor any of the Partnership Representatives shall use Confidential Information other than in connection with the business of the Partnership or a Partner's performance of its obligations under this Agreement until expiration or earlier termination of this Agreement. In addition, after the expiration or earlier termination of this Agreement, no Partner shall disclose to anyone any Confidential Information obtained by the Partner from the Partnership, a Partner or an affiliate of a Partner, other than upon the prior written consent of the party or parties providing the Confidential Information unless a Confidentiality Exception applies.

- 11.2. The restrictions set forth in Section 11.1 above shall not apply to any information which becomes publicly known through no fault of the Partners and their affiliates which received or was given access to such information by the other party or parties.
 - 12. Rights and Duties of the Partners.

12.1. Indemnification.

12.1.1. Actions by Third Persons. To the extent and in the manner permitted by the laws of the District of Columbia, the Partnership shall indemnify, defend and hold harmless any Person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, other than an action by, or in the right of the Partnership, by reason of the fact that such Person is or was a Director, trustee, officer, employee or agent of the Partnership, or is or was a Partner (including the General Partner in its role as Tax Matters Partner), or an officer, director, trustee, employee or agent of such Partner, or is or was serving at the request of the Partnership as a director, officer, employee or agent of another limited liability company, corporation, partnership, joint venture, trust or other enterprise against expenses (including but not limited to attorneys' fees, judgments, fines and amounts paid in settlement) actually and reasonably incurred by such Person in connection with such action, suit or proceeding (i) if such Person acted in good faith and in a manner such Person reasonably believed to be in, or not opposed to, the best interests of the Partnership, (ii) if the acts or omissions forming the basis for such allegation were appropriately authorized by the Partnership, and (iii) if, with respect to any criminal action or proceeding, such Person had no reasonable cause to believe the conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction or upon a plea of nolo contenders or its equivalent, shall not, of itself, create a presumption that the Person did not act in good faith and in a manner which such Person reasonably believed to be in or not opposed to, the best interests of the Partnership and, with respect to any criminal action or proceeding, had reasonable cause to believe that conduct was unlawful. Indemnification shall not be permitted if a Partner has been adjudged liable for personal benefits improperly received, willful misconduct, recklessness, or gross negligence with respect to the business of the Partnership.

12.1.2. Determination. Any indemnification under the provisions of Section 12.1.1 of this Agreement, unless ordered by a court, shall be made available to the Director, trustee, Partnership only as authorized in the specific case following a specific determination that indemnification of the Director, trustee, Partner, officer, employee or agent is proper under the circumstances because such Person has met the applicable standard of conduct set forth in Section 12.1.1 of this Agreement. Such determination shall be made:

(a) by the Partners by the vote of a majority in Percentage Interest consisting of Partners who were not parties to such action, suit or proceeding; or

(b) if the disinterested Partners so direct, by independent legal counsel in a written opinion.

- 12.1.3. Expense Advances. Expenses incurred by an officer of the Partnership or Parties in defending a civil or criminal action, suit or proceeding may be paid by the Partnership in advance of the final disposition of such action, suit or proceeding, determined on a case-by-case basis, if the action, suit or proceeding arises from the performance by such officer or Partner of Partnership duties; and upon receipt of an undertaking by or on behalf of such officer or Partner to repay such amount if it shall ultimately be determined that such officer or Partner is not entitled to be indemnified by the Partnership as authorized by the provisions of this Section 12. Such expenses incurred by other employees and agents may be so paid upon such terms and conditions, if any, as the Partners deem appropriate.
- 12.1.4. Insurance. The Partnership may, to the full extent permitted by the laws of the District of Columbia, but only to such extent as may be determined by the Partners, purchase and maintain insurance on behalf of any Person who is or was a Director, trustee, officer, employee or agent of the Partnership, or is or was a Partner or an officer, director, employee or agent of such Partner, or is or was serving at the request of the Partnership as a partner, officer, employee or agent of another limited liability company, corporation, partnership, joint venture, trust or other enterprise against any liability asserted against and incurred by such Person in any such capacity or arising out of such Person's status as such, whether or not the Partnership would have the power to indemnify such Person against such liability under the provisions of this Section 12.
- 12.1.5. Continuation of Indemnification. The indemnification and advancement of expenses provided by or granted pursuant to this Section 12 shall continue as to a Person who has ceased to be a Director, trustee, Partner, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a Person.

12.2. Partnership Business Opportunities.

12.2.1. New Business Opportunities. The Partnership shall have the first right to acquire, manage or operate any opportunity of a Partner to engage in any healthcare or related activity within the Partnership's Service Area which is not conducted by a Partner immediately after the Transfer Date. Accordingly, if a Partner is presented such an opportunity, such Partner shall offer the Partnership the first right to acquire, manage or operate such activity. An offer under this Section 12.2.1 shall be based on a summary term sheet prepared in good faith by the presenting Partner, or by the General Partner if the opportunity was first made available to the Partnership. Such term sheet shall describe the nature of the opportunity, its relationship to the activities of the Partnership, and any other material terms and conditions of the opportunity relevant to its consideration by the Partnership. The Partners intend that a business opportunity of the Partnership may be

pursued in whatever manner best accomplishes the purposes of the Partnership including, but not limited to, having ownership of assets used in the business in the name of one or more Partners. The Partnership shall determine whether to act on the business opportunity in accordance with Section 7.3(i). If the Partnership does not state in writing to the proposing Partner that it is willing to acquire, manage or operate the business opportunity within thirty (30) days after the Partnership's receipt of the offer, or if the Partnership does not proceed with all due diligence to consummate the transaction, either Partner may negotiate for and acquire, manage or operate such business opportunity.

- 12.2.2. Remedies. In the event of an actual or threatened breach by the Partnership or a Partner of this Section 12.2 the nonbreaching Partner or its affected affiliate shall be entitled to an injunction restraining the breaching party from the prohibited conduct. Notwithstanding any other provision of this Agreement, if a court of competent jurisdiction should hold that the scope (geographic or otherwise) of the foregoing covenants is unreasonable, then, to the extent permitted by law, the court may prescribe a scope (geographic or otherwise) that is reasonable and judicially enforceable. Nothing herein stated shall be construed as prohibiting a nonbreaching Partner or any affiliate thereof from pursuing any other remedies available to it for such breach or threatened breach, including, the recovery of damages from any breaching party.
- 12.3. Covenant Not to Compete. As of the Transfer Date, the Partners and their affiliates shall not, without the prior written consent of the Partnership, own or operate, directly or indirectly, any healthcare related business which may be competitive with the activities of the Partnership within the Partnership's Service Area; provided, however, that this Section 12.3 shall not apply to (i) the operations of the Partners and their respective affiliates in place as of the Effective Date of this Agreement that are listed on Exhibit 12.3 hereto or (ii) the particular opportunities presented to the Partnership under Section 12.2.1, which the Partnership has chosen not to pursue. To the extent that any portion of the provisions of this Section 12.3 shall be deemed by a court of competent jurisdiction to exceed the time or geographic limits or any other limitations permitted by applicable law, then this Section 12.3 shall be deemed reformed to the maximum extent permitted by applicable law.
- 12.4. Partnership Assets. The credit and assets of the Partnership shall be used solely for the benefit of the Partnership and shall not otherwise be used to further the personal gain of any of the Partners. Title to and ownership of all of the assets of the Partnership shall at all times be vested in and stand in the name of the Partnership. The General Partner shall execute, file and record such documents which may become necessary to reflect the Partnership's ownership of such property in such public offices as may be required.
- 12.5 MFA Activities. The provisions of Sections 12.2 and 12.3 shall not apply to activities of the Medical Faculty Associates relating to the type of health care services provided by the members of the Medical Faculty Associates in their respective offices as of the Effective Date.

- 13. Transfer of Partnership Interests.
 - 13.1. Restrictions on Transfer.
- 13.1.1. Restrictions on Transfers by UHS and the University. Neither UHS nor the University may Transfer their respective Interests in the Partnership unless the Partner desiring to Transfer its Interest under this Section 13.1 obtains the prior written approval of the nontransferring Partner, which approval shall not be unreasonably withheld.
- 13.1.2. Limitations on Rights of Assignees. Transfers of interests in the Partnership under this Section 13.1 shall be, and any Person(s) acquiring a Transferred Interest (an "Assignee") shall acquire such Transferred Interest in the Partnership, subject to all of the terms and conditions of this Agreement. A Transfer of an Interest in the Partnership shall not relieve the transferring Partner of its duties and obligations to the Partnership unless the General Partner agrees in writing to release the transferring Partner. An Assignee shall not be admitted as a Partner (a "Substitute Partner") unless all the requirements of Section 13.3 of this Agreement are satisfied. Absent admission as a Substitute Partner, Assignees shall be permitted to receive only the share of the Partnership's income, gain, deductions, credits and losses to which the transferring Partner was previously entitled.
- 13.2. New Partners. New Partners may be admitted to the Partnership upon satisfaction of the requirements of Sections 3.2 and 13.3 of this Agreement and subject to the provisions of Section 7.3. As new Partners are admitted, any adjustment in the Percentage Interests of existing Partners shall be made as mutually agreed.
- 13.3. Substitution of Assignees and Admission of New Partners. Subject to Section 13.1 hereof relating to Transfers of Interests and Section 13.2 hereof relating to the admission of new Partners, an Assignee may become a Substitute Partner with all the rights and liabilities of any Partner under this Agreement, and new Partners may be admitted to Partnership, if and only if (i) the Assignee or new Partner executes and agrees to be bound by this Agreement in the place of the transferring Partner or as a new Partner, as applicable; (ii) in the event such Assignee is a corporation, partnership (general or limited), trust or limited liability company, such Assignee or new Partner provides the General Partner with evidence satisfactory to counsel for the Partnership of such Assignee's or new Partner's authority to become a Partner under the terms and conditions of this Agreement; (iii) the assignment or admission instruments, documents or statements, if any, are prepared, executed, acknowledged, filed, published and delivered as required by the Act or otherwise; (iv) the Assignee or new Partner pays or obligates itself to pay any and all reasonable costs and expenses, including attorneys' fees, incurred by the Partnership in connection with such substitution or admission; and (v) the Partners and the Partnership Board approve the admission of the new Partner, or the Assignee as a Substitute Partner, under Section 7.3(c) of this Agreement, which approval shall not be unreasonably withheld. The admission of such Assignee or new Partner shall not cause a dissolution of the Partnership. In addition, this

Agreement and Exhibit 1.34 hereto shall be amended to reflect the admission of new Partners and reflect the reallocation of the existing Partners' respective Percentage Interests in the Partnership without the necessity of any existing Partner's signature.

- 13.4. Effective Date of Assignment or Admission. For the purpose of allocating Partnership income, gains, deductions, credits and losses, an Assignee or new Partner shall be treated as a Partner on such date as consent of the nontransferring Partners to such Transfer as is required under this Section 13 is obtained.
 - 14. Dissolution, Liquidation and Termination of the Partnership.
- 14.1. Limitations. The Partnership may be dissolved, liquidated and terminated only pursuant to the provisions of this Section 14. The Partners hereby irrevocably waive any and all other rights they may have to cause a dissolution of the Partnership or a sale or partition of any or all of its assets.
- 14.2. Exclusive Causes. The following events, and only the following events, shall cause the Partnership to be dissolved, liquidated and terminated pursuant to the provisions of this Section 14.2:
- 14.2.1. Expiration. The expiration of the Term of the Partnership.
- $\ensuremath{\text{14.2.2.}}$ Consent. The consent of all Partners to dissolve the Partnership.
- 14.2.3. Unlawful Event. The occurrence of any event which makes it unlawful for the business of the Partnership to be carried on, whether by the Partners or the Partnership.
- 14.2.4. Sale of All or Substantially All of Partnership Assets. The sale of all or substantially all of the assets of the Partnership and the receipt of all cash proceeds from such sale.
- 14.2.5. Failure to Execute and Close. The failure of any of the parties to this Agreement to execute any of the Ancillary Agreements or to close the transactions contemplated by Section 3 of this Agreement by December 31, 1997 or by such later date mutually agreed to by the parties.
- 14.2.6. Expiration or Termination of Ground Lease. The termination of the Ground Lease between the University and the Partnership due to expiration or a breach by the University, unless otherwise approved by the Partners.

- \$14.2.7.\$ Bankruptcy of a Partner. The Bankruptcy of any Partner unless the Partnership is continued by the consent of the other Partner(s).
- $$\tt 14.2.8.$ Bankruptcy of the Partnership. The Bankruptcy of the Partnership.
- 14.2.9. Judicial Dissolution. A decree of dissolution rendered by a court of competent jurisdiction pursuant to District of Columbia Code Section 41-482.
- 14.2.10. Breach by a Partner. A material breach of this Agreement by any Partner; provided, however, that the Partner alleged to have breached this Agreement shall have twenty (20) business days to cure such alleged breach to the satisfaction of a majority in Percentage Interest of the nonbreaching Partners.
- 14.3. Liquidating Trustee; Continuation of Business. Upon the dissolution of the Partnership, the General Partner shall act as the "Liquidating Trustee" of the Partnership. During the period of the dissolution, liquidation and termination of the Partnership pursuant to the provisions of this Section 14, the business of the Partnership may be continued by the Liquidating Trustee to the extent necessary to allow an orderly winding up of the Partnership's affairs, including without limitation, the liquidation of the Partnership pursuant to the provisions of Section 14.4 below.
- 14.4. Liquidation of Partnership. Upon the dissolution of the Partnership pursuant to the provisions of Section 14.2 and within a reasonable time thereafter, the . Liquidating Trustee shall wind up the Partnership's business and affairs in the following manner.
- 14.4.1. The Liquidating Trustee shall obtain and furnish an accounting with respect to all Partnership accounts and the Capital Accounts of each Partner and with respect to the Partnership's assets and liabilities and its operations from the date of the last financial statements of the Partnership to the date of its liquidation.
- 14.4.2. To the extent the Liquidating Trustee deems appropriate, all material, equipment, and real and personal property of the Partnership of any kind or nature may be sold.
- 14.4.3. The Liquidating Trustee shall pay the expenses of liquidation and the debts of the Partnership from the Partnership's assets, including debts owing to the Partners in the order of priority provided by law, except the claims of secured creditors whose obligations will be assumed or otherwise transferred upon the liquidation or Distribution of the Partnership's assets.

- 14.4.4. The Liquidating Trustee shall ascertain the fair market value by appraisal or other reasonable means of all assets of the Partnership not sold and intended to be distributed to the Partners in connection with the liquidation, and each Partner's Capital Account shall be charged or credited, as the case may be, as if such properly had been sold at such fair market value and the gain or loss realized thereby had been allocated to and among the Partners.
- 14.4.5. Remaining proceeds shall be paid to the Partners who have net positive balances in their Capital Accounts, as determined after taking into account all Capital Account adjustments for the Partnership's Fiscal Year, until all such balances have been reduced to zero or, in the event proceeds are insufficient, pro rata on account thereof.
- 14.4.6. In the event any proceeds remain after Distributions pursuant to Sections 14.4.1 through 14.4.5 above, such proceeds shall be distributed to the Partners according to their respective Percentage Interests.

15. Miscellaneous.

- 15.1. Notices. Any and all notices or demands in connection with this Agreement shall be in writing and served either personally, by certified mail, return receipt requested, or by Federal Express or other reputable overnight courier, at the respective addresses set forth opposite the signatures of the Partners, or to such other addresses as any of them shall from time to time designate. If served personally, service shall be conclusively deemed made at the time of such service. If served by certified mail, service shall be conclusively deemed made on the fourth business day after the deposit thereof in the United States mail, postage prepaid, addressed pursuant to the provisions of this Section 15.1. If served by Federal Express or other overnight reputable courier, service shall be conclusively deemed made on the day after transmittal thereof.
- 15.2. Hospital Name. Pursuant to the Trademark License Agreement between the University and the Partnership, the Partnership shall continue to use the name "The George Washington University Hospital" in connection with the Hospital so long as it continues to operate the Hospital in the Washington Circle area of the District of Columbia unless such name is required to be changed to comply with applicable law. The University agrees that it shall not license the name "The George Washington University Hospital" to, or use the name in connection with, any entity, facility, or service without the consent of the Partnership.

- 15.3. Attorneys' Fees. Should any party retain counsel to enforce any of the provisions herein or protect its interest in any matter arising under this Agreement, or to recover damages by reason of any alleged breach of any provision of this Agreement, the losing party in any action pursued in a court of competent jurisdiction (the formality of which is not legally contested) shall pay to the prevailing party all costs, damages, and expenses incurred by the prevailing party, including, without limitation, attorneys' fees and costs incurred in connection therewith.
- 15.4. Interpretation. No provision of this Agreement is to be interpreted for or against any party because that party or that party's legal representative drafted such provision.
- 15.5. Waiver. No breach of any provision of this Agreement may be, waived except in writing. Waiver of any one breach of any provision of this Agreement shall not be deemed to be a waiver of any other breach of the same or any other provision of this Agreement.
- 15.6. Severability. In the event that any covenant, condition or other provision contained in this Agreement is held to be invalid, void or illegal by any court of competent jurisdiction, the same shall be deemed severable from the remainder of this Agreement and shall in no way affect, impair or invalidate any other covenant, condition or other provision contained in this Agreement so long as severance of the invalid provision does not materially alter the rights and obligations of the parties. If such condition, covenant or other provision shall be deemed invalid due to its scope or breadth, such covenant, condition or other provision shall be deemed valid to the extent of the scope or breadth permitted by law.
- 15.7. Additional Documents. In addition to the documents and instruments to be delivered as provided in this Agreement, each of the parties shall, from time to time at the request of the other parties, execute and deliver to the other parties or the Partnership such other documents and shall take such other action as may be reasonably required to carry out more effectively the terms of this Agreement.
- 15.8. Governing Law. This Agreement shall be governed by, and construed in accordance with, the laws of the District of Columbia.
- 15.9. Headings. Paragraph titles or captions contained in this Agreement are inserted as a matter of convenience and for reference purposes only, and in no way define, limit, interpret, extend or describe the scope of this Agreement or any provision of this Agreement.
- 15.10. Survival of Obligations. All indemnities, and continuing agreements and covenants contained or referred to in this Agreement shall survive the execution and delivery

of this Agreement, and it shall not be a condition precedent to any indemnity set forth herein that the indemnified party shall have made any payment on account of any claim, loss, damage, obligation, liability, deficiency, penalty, cost or expense indemnified against herein.

- 15.11. Amendment. The Partners acknowledge that the exhibits to this Agreement will be finalized subsequent to the execution of this Agreement. Subject to the provisions of Section 7.3, this Agreement may be amended by the Partnership Board.
- 15.12. Successors and Assigns. This Agreement shall be binding upon and shall inure to the benefit of the parties and their respective representatives, successors and permitted assigns.
- 15.13. Third Party Beneficiaries. Nothing in this Agreement shall be for the benefit of anyone not a party to this Agreement.
- 15.14. Incorporation. All exhibits attached to this Agreement are incorporated herein by this reference.
- 15.15. Execution in Counterparts. This Agreement may be executed in two (2) or more counterparts, each of which shall be an original instrument, but all of which shall constitute one and the same agreement.
- 15.16. Entire Agreement. This Agreement, including any exhibits, supersedes all previous written or oral agreements between the parties hereto with respect to the subject matter hereof. Furthermore, upon the attachment of the last exhibit to be attached to this Agreement, the Letter of Intent, dated March 25, 1997, between the University and Universal Health Services, Inc., shall be terminated and the parties thereto shall have no further rights or obligations thereunder.

UHS of D.C., INC

367 South Gulph Road P.O. Box 61558

King of Prussia, PA 19046-0958

By: [SIG]

Richard C. Wright

Its:

Vice President

THE GEORGE WASHINGTON UNIVERSITY

2121 I Street, N.W. Suite 701, Rice Hall Washington, DC 20052 By: [SIG]

Louis H. Katz

Its: Vice President and Treasurer

-35-

Exhibit 1.35

PERCENTAGE INTERESTS OF THE PARTNERS

GENERAL PARTNER

Name Interest

UHS 1%

LIMITED PARTNERS

Name Interest

UHS 79%

The University 20%

Exhibit 7.2.3

INITIAL BOARD OF DIRECTORS

-37-

AMENDMENT NO. 2 TO POOLING AGREEMENT

AMENDMENT NO. 2, dated as of April 17, 1997 (this "Agreement"), to Pooling Agreement dated as of November 16, 1993 (as amended, supplemented or otherwise modified from time to time, the "Pooling Agreement"), among UHS Receivables Corp., a Delaware corporation (together with its successors and assigns, "Finco"), Sheffield Receivables Corporation, a Delaware corporation (together with its successors and assigns, "Sheffield"), and First Bank National Association, a national banking association (successor in interest to Continental Bank, National Association), as trustee (in such capacity, together with its successors and assigns, the "Trustee").

WITNESSETH:

WHEREAS, pursuant to the Sale and Servicing Agreements (capitalized terms used without definition in the recitals have the meanings assigned to them in the Pooling Agreement, including without limitation definitions incorporated by reference from the Definitions List) the Hospitals have agreed to sell, and Finco has agreed to purchase, the Receivables and other Transferred Property;

WHEREAS, Finco has assigned and transferred to the Trustee on behalf of the Trust and pursuant to the terms and conditions set forth in the Pooling Agreement all of its right, title and interest in, to and under the Receivables and other Finco Transferred Property;

WHEREAS, Finco, Sheffield and the Trustee wish to amend the Pooling Agreement to provide for Finco to transfer to the Trustee Receivables from additional Hospitals entering into Sale and Servicing Agreements as of the date hereof, to reflect their agreement not to create Trade Receivables Investment Participation (TRIPS) as originally contemplated, and to make certain other changes;

NOW THEREFORE, the parties hereto agree as follows:

Section 1. Additional Hospitals. The Pooling Agreement is hereby amended to provide that the receivables and other property of the additional hospitals entering Sale and Servicing Agreements on or after the date hereof shall be transferred to the Trustee pursuant to the Pooling Agreement; such additional hospitals shall be deemed "Hospitals" for all purposes under the Pooling Agreement; and all references in the Pooling Agreement to "Receivables", "Transferred Property" and other terms shall be deemed to include the analogous terms relating to such hospitals.

Section 2. Procedures for Adding Hospitals. Additional hospitals may be added after written notice from Finco to Sheffield and the Trustee upon receipt by Finco and the Trustee of written consent by Sheffield to the addition of such hospital and the execution and delivery of a Sale and Servicing Agreement by such hospital and Finco.

Section 3. Termination of TRIPS Provisions. The parties have agreed not to create the TRIPS Participations contemplated at the time the Pooling Agreement was executed. Accordingly, the parties hereby agree that:

- (a) the Trustee shall have no obligation to maintain a TRIPS Register or provide notices to any TRIPS Holder, and the Pooling Agreement is hereby amended to delete the requirement that the Trustee maintain a TRIPS Register or provide notices to any TRIPS Holder; and
- (b) the Trustee shall have no obligation to maintain the TRIPS Payment Account, the TRIPS Sub-account and the TRIPS Interest Sub-account, and the Pooling Agreement is hereby amended to delete the requirement that the Trustee maintain such accounts.

Section 4. Other Amendments.

- (a) The Pooling Agreement is hereby amended to replace the definitions of "Concentration Limit", "Offset Reserves", "Maximum Aggregate Capital", "Maximum Sheffield Capital" and "Preferred Insurer/HMO/PPO" incorporated by reference from the Definitions List with the definitions set forth on Exhibit A to this Amendment.
- (b) Sections 2.13(b) of the Pooling Agreement is hereby amended to increase the maximum Aggregate Capital from \$85,000,000 to \$100,000,000.
- (c) The Trustee shall have no obligation to maintain the Sheffield Sub-account. All reference in the Pooling Agreement to such account shall be deemed to be references to the Collateral Account.
- Section 5. Effect of Amendment. The Pooling Agreement as modified by this Amendment shall remain in full force and effect, and all references in the Pooling Agreement to "this Agreement" shall mean the Pooling Agreement as amended hereby unless the meaning requires otherwise.
- Section 6. Counterparts. This Amendment may be executed in any number of copies, and by the different parties hereto on the same or separate counterparts, each of which shall be deemed to be an original instrument.
- Section 7. GOVERNING LAW. THIS AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE INTERNAL LAW OF THE STATE OF NEW YORK WITHOUT REFERENCE TO CONFLICTS OF LAW RULES OF THE STATE OF NEW YORK.

3

IN WITNESS WHEREOF, Finco, Sheffield and the Trustee have caused this Agreement to be duly executed by their duly authorized officers, all on the day and year first above written.

UHS RECEIVABLES CORP.

SHEFFIELD RECEIVABLES CORPORATION

By: /s/ [SIG]

Title: Director

FIRST BANK NATIONAL ASSOCIATION, Trustee

By: /s/ [SIG]

Title: Vice President

4 Acknowledged and Agreed:

BARCLAYS BANK PLC, as Liquidity Agent

By: /s/ [SIG]

Title: Director

NEW DEFINITIONS

Concentration Limit: On any date of determination and when used with reference to Eligible Receivables of a specified type held by Finco and in which the Participants have acquired the Participations, the following applicable percentage of the aggregate Outstanding Balance of all Eligible Receivables so held by Finco represented by such type of Eligible Receivables, after giving effect to any purchases by Finco of Eligible Receivables to be effected on such date:

Obligor

Medicare Medicaid Preferred Blue Cross/Blue Shield Aggregate Blue Cross/Blue Shield Aggregate Insurers/HMOs/PPOs Champus Worker's Compensation Preferred Insurer/HMO/PPO Other Blue Cross/Blue Shield	40% 15% 10% 10% 60% 7% 10% 7%
Other Insurer/HMO/PPO	3.75%

Offset Reserves: On any date, the sum of 9(a) \$1,000,000, (b) an amount determined as of the last day of the preceding Settlement Period to be equal to the unpaid portion, as reflected in all audited periodic cost reports filed by all Hospitals with the appropriate state and federal Governmental Authorities under the applicable Medicaid programs and with HCFA under Medicare, of the net amount payable by the aggregate of all Hospitals for which cost reports indicate amounts payable to such governmental authorities and (c) an additional amount, if greater than zero, determined as of the date of completion of all audits of all periodic cost reports filed during the preceding fiscal year of UHS for all UHS Entities to be equal to (i) the greater of (A) 1.5 times the highest amount (rounded to the nearest \$1,000,000) payable to such Governmental Authorities and reflected in the past three annual audited cost reports of all Hospitals for which audits determined net amounts to be payable to such Governmental Authorities and (B) three times the amount payable to such Governmental Authorities and reflected in the most recent audited cost reports of all Hospitals for which audits determined net amounts to be payable to such governmental authorities minus (ii) an amount equal to the aggregate Collections received in respect of Self-Pay Receivables during the period of two Consecutive Settlement Periods occurring in the fiscal year immediately preceding such date of determination for which such aggregate Collections were the lowest.

MAXIMUM AGGREGATE CAPITAL: \$100,000,000.

MAXIMUM SHEFFIELD CAPITAL: \$75,000,000, as such amount may be increased from time to time in accordance with Section 2.13 of the Pooling Agreement.

PREFERRED INSURER/HMO/PPO: At any time of reference, individually, any Obligor which is a commercial insurer, health maintenance organization, preferred provider organization or similar entity (excluding Blue Cross/Blue Shield) and which is rated at least BBB- by S&P and Baa-3 by Moody's.

```
3-M0S
       DEC-31-1997
          JAN-01-1997
            MAR-31-1997
                   1
                         2,281
               154,952
                    0
                  21,915
            203,371
                      870,620
              286,761
989,963
       161,184
                      258,693
             0
                        0
                        323
                  477,796
989,963
                            0
            340,170
                              0
               249,421
             28,149
            23,663
            4,956
              33,981
                 12,451
          21,530
                    0
                    0
                          0
                 21,530
$0.65
                 $0.65
```